Author’s response to reviews

Title: Dialectical Behaviour Therapy for treating adults and adolescents with emotional and behavioural dysregulation; study protocol of a coordinated implementation in a publicly funded health service.

Authors:

Daniel Flynn (danielm.flynn@hse.ie)
Mary Kells (mary.kells@hse.ie)
Mary Joyce (marychjoyce@gmail.com;maryc.joyce@hse.ie)
Catalina Suarez (csuarezluque@hotmail.com)
Conall Gillespie (conall.gillespie@hse.ie)

Version: 1 Date: 16 Aug 2017

Author’s response to reviews:

Comment:

1. Ethical and Funding Approval Documentation

Before we can proceed with your submission, can you please forward copies of all ethical approval and funding approval for our records. These documents should be sent as email attachments to the following email address, BMCSeriesEditorial@biomedcentral.com. Please DO NOT upload these documents as additional files in the submission system. If your documents are not in English, please could you provide translated versions of the relevant parts. These should be endorsed and signed by a contactable person at the institution. Please also include the original documents. Can you please confirm in your cover letter that you have forwarded the requested documentation to either BMCSeriesEditorial@biomedcentral.com or Carlo.RyeChua@springer.com.

Response:

I confirm that I have sent copies of all ethical approval for your records to Carlo.RyeChua@springer.com on August 15th 2017.

Comment:

2. Funding
A study is considered to be externally funded if the authors have been awarded a grant for the study by a major funding body (e.g. governmental funding/award from a charitable foundation). If a study has not received external funding, then the study protocol will be sent for peer-review with a member of our Editorial Board. If a study has received funding/assistance from a commercial organization, this should be clearly stated in the 'competing interests' section of your manuscript, and the study protocol will be sent for peer-review by a member of our Editorial Board. Can you please confirm whether your study protocol has undergone peer-review by the funding body.

Response:

This study is funded by the National Office for Suicide Prevention in the Health Service Executive, Ireland. Funding was granted to train 16 DBT teams in Community Mental Health Services across the Republic of Ireland over a two-year period and to evaluate this implementation initiative. This study protocol did not undergo peer-review by the funding body.

Comment:

3. Consent to participate

Please include details of the procedures for obtaining informed consent to participate in your 'Ethics and consent to participate' declaration.

Response:

The procedures for obtaining informed consent to participate are now detailed in the 'Ethics and consent to participate' declaration.

Comment:

4. Authors’ contributions:

Please note that based on the current authors’ contributions section, several listed authors (GC) do not automatically qualify for authorship. Acquisition of funding, collection of data, writing of the manuscript or general supervision of the research group, alone, does not usually justify authorship.

We ask that you ensure that author contributions are in line with the ICMJE guidelines (below), and that all listed authors have performed all four points specified below.

Response:
We have reviewed the ICMJE guidelines regarding author contributions and confirm that all authors fulfil these criteria. This is not clarified in the 'Authors' contributions' section of the revised manuscript.